Author's response to reviews

Title: Serum procalcitonin and CRP levels in non-alcoholic fatty liver disease: a case control study

Authors:

Nevin Oruc (nevintr@yahoo.com)
Omer Ozutemiz (omer.ozutemiz@ege.edu.tr)
Gul Yuce (gulyuce@ege.edu.tr)
Ulus S Akarca (ulus.akarca@ege.edu.tr)
Galip Ersoz (galip.ersoz@ege.edu.tr)
Fulya Gunsar (fulya.gunsar@ege.edu.tr)
Yucel Batur (yucelbatur@yahoo.com)

Version: 4 Date: 1 February 2009

Author's response to reviews:

Dear editors,

We would like to thank you for accepting our manuscript. We corrected and proof read the final manuscript.

1. Author names and emails corrected as in the submission system
2. All tables formatted according to the journal style.
3. Abbreviation (NAFLD) was used instead of non alcoholic fatty liver disease in introduction final paragraph.
4. Typing errors were corrected.
5. In results, typing error in CRP SEM level was corrected
6. In methods typing error p<0,005 corrected as p<0,05.
7. In table and text HOMO-R changed as HOMA-IR
8. BMI is added as an abbreviation

Yours sincerely